Author’s response to reviews

Title: Cardiac Valvular Abnormalities Associated with Use and Cumulative Exposure of Cabergoline for Hyperprolactinemia: The CATCH Study

Authors:

Amer Budayr (drbudayr.endocrine@gmail.com)
Thida Tan (thida.x.tan@kp.org)
Joan Lo (joan.c.lo@kp.org)
Jonathan Zaroff (jonathan.g.zaroff@kp.org)
Grace Tabada (grace.h.tabada@kp.org)
Jingrong Yang (jingrong.yang@kp.org)
Alan Go (alan.s.go@kp.org)

Version: 4 Date: 11 Feb 2020

Author’s response to reviews:

Dear Ms. Debney and Dr. Glynn:

Thank you for the provisional acceptance for our manuscript, and I am confirming that all of the requested remaining revisions have been made. Per your instructions, these include:

1. Reformat manuscript to BMC Endocrine Disorders submission guidelines.

This has been done.

2. Availability of data and materials section.

We have confirmed the accuracy of the description of our statement regarding the lack of availability of study data in the Declarations section (pg 15).

3. Funding section.

We have confirmed that the sponsor had no role in the design of the study, data collection, data analysis, interpretation of data, the decision the submit results for publication, or in the preparation, review or approval of the manuscript. This is updated in the Declaration section (pg 15).
4. Remove any additional unneeded files.
This has been done.

5. If the manuscript text is duplicated in the manuscript file, please remove the duplicate text.
Not applicable.

6. Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
This has been done.

I hope that we have now addressed any outstanding issues, and we have double-checked the entire final version of the manuscript again for accuracy and clarity. Please let me know if there are any questions or additional information needed.

Thank you very much.

Best regards,
Alan S. Go, MD